Author’s response to reviews

Title: Absolute risk representation in cardiovascular disease prevention: comprehension and preferences of health care consumers and general practitioners involved in a focus group study

Authors:

Sophie Hill (sophie.hill@latrobe.edu.au)
Janet Spink (jdspink@bigpond.net.au)
Dominique Cadilhac (dcadilhac@nsri.org.au)
Adrian Edwards (edwardsag@cardiff.ac.uk)
Caroline Kaufman (cakaufman@seton.org)
Sophie Rogers (srogers@unimelb.edu.au)
Rebecca Ryan (r.ryan@latrobe.edu.au)
Andrew Tonkin (andrew.tonkin@med.monash.edu.au)

Version: 4 Date: 11 February 2010

Author’s response to reviews:

For this submission, we are responding to the final editing requests from the publisher. These are itemised in our response to the email sent to us by the Production Team.